Author's response to reviews

Title: A prospective study of weight development and behavior problems in toddlers: the Norwegian Mother and Child Cohort Study

Authors:

Susan Garthus-Niegel (susanniegel@gmail.com)
Knut A. Hagtvet (k.a.hagtvet@psykologi.uio.no)
Margarete E. Vollrath (margareteerika.vollrath@fhi.no)

Version: 5 Date: 18 October 2010

Author's response to reviews:

Dear BMC Series Editorial Production Team,

Please see our revised manuscript with the following formatting changes:

1. The tables have been included in the manuscript in a tables section following the references.
   The vertical text has been removed and been rewritten horizontally.

2. Line numbers have been removed.

3. All highlighting and tracking has been removed from the manuscript.

4. All text is displayed in black throughout the manuscript.

5. The full citation details for references 9, 12 and 15 are now provided.

6. We have now removed the title and figure number from the image files.